

Summary of the Peer-Review Report:

Comment:

This article is densely written with complex sentences and medical terminology; please consider simplifying the terminology as much as possible or providing brief descriptions for highly technical terms.

Answer: Terminology has been simplified

Comment:

Lacks a patient-centred perspective. More discussion of long-term results should be included or suggestions for areas for future research would enhance the practical usefulness of the article

Answer: Future Research Section added:

Post POEM reflux can have health-related and economic implications in the long term. Therefore, long-term outcomes of post-POEM reflux, its causes, prevention, and management options can emerge as arenas for potential future research.

Comment

The World Journal of Gastroenterology article which this editorial discussed has not been listed in the references list

Answer : article has been cited in new manuscript